Author's response to reviews

Title: A genomic copy number variant analysis implicates the MBD5 and HNRNPU genes in Chinese children with infantile spasms and expands the clinical spectrum of 2q23.1 deletion

Authors:

Xiaonan Du (dresmedu@126.com)
Yu AN (ay.fudan@gmail.com)
Lifei Yu (yulifei7711@163.com)
Renchao Liu (lrc360681@gmail.com)
Yanrong Qin (gingyanrong521@163.com)
Xiaohong Guo (xhguo@shmu.edu.cn)
Daokai Sun (daokaisun@163.com)
Shuizhen Zhou (szzhou@shmu.edu.cn)
Bai-lin Wu (bai-lin.wu@childrens.harvard.edu)
yong-hui Jiang (yong-hui.jiang@duke.edu)
Yi Wang (yiwang@shmu.edu.cn)

Version: 6 Date: 12 May 2014

Author's response to reviews: see over
Dear Editor

We are very appreciated to the accepting of our manuscript. We have revised the manuscript according to the editorial requests. We hope you will find the revised manuscript is ready for publication.

Sincerely,

Wang Yi, MD. PhD.
Professor and Vice President
Division of Neurology
Children's Hospital of Fudan University
399 Wan Yuan Road, Shanghai City 20032, P.R. China
E-mail: yiwang@shmu.edu.cn
Point-by-Point Response

1. Please include the email addresses of all co-authors in the title page.

   The email addresses have been added in red.

2. Please edit the 'Authors' contributions' section so that all listed authors fulfill the criteria for authorship according to ICJME guidelines. For further information, please see here: http://www.biomedcentral.com/bmcmedgenet/authors/instructions/researcharticle#formatting-contributions

   All the authors listed meet the criteria.

3. Please remove additional file 1 and include these tables after the references section.

   The tables are correctly located now.

4. Please state (in the methods section of the manuscript) whether parents gave their consent for the publication of the photographs in Figures 1-3. If not, then please remove these photographs from the figures.

   We have states the consent in method in red. For detail, S100 and S162, they agreed with use of their photographs, and for S37, they want to blacken the eyes.